\begin{thebibliography}{10}

\bibitem{dash75}
J.~G. Dash, {\em Films on Solid Surfaces} (Academic, New York, 1975).

\bibitem{schick80}
M. Schick,  in {\em Phase Transitions in Surface Films}, edited by J.~G. Dash
  and J. Ruvalds (Plenum, New York, 1980).

\bibitem{bruch97}
L.~W. Bruch, M.~W. Cole, and E. Zaremba, {\em Physical Adsorption: Forces and
  Phenomena} (Oxford, New York, 1997).

\bibitem{bretz77}
M. Bretz, Phys. Rev. Lett. {\bf 38},  501  (1977).

\bibitem{greywall93}
D.~S. Greywall, Phys. Rev. B {\bf 47},  309  (1993).

\bibitem{nielsen80}
M. Nielsen, J.~P. McTague, and L. Passell,  in {\em Phase Transitions in
  Surface Films}, edited by J.~G. Dash and J. Ruvalds (Plenum, New York, 1980).

\bibitem{abraham87}
F.~F. Abraham and J.~Q. Broughton, Phys. Rev. Lett. {\bf 59},  64  (1987).

\bibitem{carneiro76}
K. Carneiro {\it et~al.}, Phys. Rev. Lett. {\bf 37},  1695  (1976).

\bibitem{hering76}
S.~V. Hering, S.~W.~V. Sciver, and O.~E. Vilches, J.\ Low Temp.\ Phys. {\bf
  25},  793  (1976).

\bibitem{rabedeau89}
T.~A. Rabedeau, Phys. Rev. B {\bf 39},  9643  (1989).

\bibitem{ecke85}
R.~E. Ecke, Q.-S. Shu, T.~S. Sullivan, and O.~E. Vilches, Phys. Rev. B {\bf
  31},  448  (1985).

\bibitem{greywall91}
D.~S. Greywall and P.~A. Busch, Phys. Rev. Lett. {\bf 67},  3535  (1991).

\bibitem{chen94}
J.~M. Mochel and M.-T. Chen, Physica B {\bf 197},  278  (1994).

\bibitem{boninsegni98}
M. Boninsegni and M.~W. Cole, J.\ Low Temp.\ Phys. {\bf 113},  393  (1998).

\bibitem{cowan98}
J. Ny\'{e}ki, R. Ray, B. Cowan, and J. Saunders, Phys. Rev. Lett. {\bf 81},
  152  (1998).

\bibitem{bretz73}
M. Bretz, Phys. Rev. Lett. {\bf 31},  1447  (1973).

\bibitem{berker78}
A.~N. Berker, S. Ostlund, and F.~A. Putnam, Phys. Rev. B {\bf 17},  3650
  (1978).

\bibitem{schick77}
M. Schick, J.~S. Walker, and M. Wortis, Phys. Rev. B {\bf 16},  2205  (1977).

\bibitem{tejwani80}
M.~J. Tejwani, O. Ferrerira, and O.~E. Vilches, Phys. Rev. Lett. {\bf 44},  152
   (1980).

\bibitem{crary87}
S.~B. Crary and D.~A. Fahey, Phys. Rev. B {\bf 35},  2102  (1987).

\bibitem{bruch93}
J.~M. Gottlieb and L.~W. Bruch, Phys. Rev. B {\bf 48},  3943  (1993).

\bibitem{reppy94}
P.~A. Crowell and J.~D. Reppy, Physica B {\bf 187},  278  (1994).

\bibitem{reppy96}
P.~A. Crowell and J.~D. Reppy, Phys. Rev. B {\bf 53},  2701  (1996).

\bibitem{mohandas95}
P. Mohandas {\it et~al.}, J.\ Low Temp.\ Phys. {\bf 101},  481  (1995).

\bibitem{pierce98}
M. Pierce and E. Manousakis, Phys. Rev. Lett. {\bf 81},  156  (1998).

\bibitem{cole80}
W.~E. Carlos and M.~W. Cole, Surf. Sci. {\bf 91},  339  (1980).

\bibitem{whitlock98}
P.~A. Whitlock, G.~V. Chester, and B. Krishnamachari, Phys. Rev. B {\bf 58},
  8704  (1998).

\bibitem{ceprev}
D.~M. Ceperley, Rev. Mod. Phys. {\bf 67},  279  (1995).

\bibitem{pierce99a}
M. Pierce and E. Manousakis, Phys. Rev. B {\bf 59},  3802  (1999).

\bibitem{imada84}
M. Takahashi and M. Imada, J. Phys. Soc. Jpn {\bf 53},  3765  (1984).

\bibitem{aziz92}
R.~A. Aziz {\it et~al.}, Mol. Phys. {\bf 77},  321  (1992).

\bibitem{bruch82}
L.~W. Bruch, Surf. Sci. {\bf 125},  194  (1982).

\bibitem{mclachlan64}
A.~D. McLachlan, Mol. Phys. {\bf 7},  381  (1964).

\bibitem{aziz79}
R.~A. Aziz {\it et~al.}, J. Chem. Phys. {\bf 70},  4330  (1979).

\bibitem{whitlock88}
P.~A. Whitlock, G.~V. Chester, and M.~H. Kalos, Phys. Rev. B {\bf 38},  2418
  (1988).

\bibitem{giorgini96}
S. Giorgini, J. Boronat, and J. Casulleras, Phys. Rev. B {\bf 54},  6099
  (1996).

\bibitem{clements93b}
B.~E. Clements, J.~L. Epstein, E. Krotcheck, and M. Saarela, Phys. Rev. B {\bf
  48},  7450  (1993).

\bibitem{gordillo98}
M.~C. Gordillo and D.~M. Ceperley, Phys. Rev. B {\bf 58},  6447  (1998).

\bibitem{bretz78}
S.~E. Polanco and M. Bretz, Phys. Rev. B {\bf 17},  151  (1978).

\bibitem{carneiro81}
K. Carneiro, L. Passell, W. Thomlinson, and H. Taub, Phys. Rev. B {\bf 24},
  1170  (1981).

\bibitem{pierce99b}
M. Pierce and E. Manousakis, editorially approved for publication in Phys. Rev.
  Lett.

\bibitem{cole81}
M.~W. Cole and D.~L. Goodstein, Rev. Mod. Phys. {\bf 53},  199  (1981).

\bibitem{derry79}
G.~D. Derry, D. Wesner, W.~E. Carlos, and D.~R. Frankl, Surf. Sci. {\bf 87},
  629  (1979).

\bibitem{carlos80}
W.~E. Carlos and M.~W. Cole, Phys. Rev. B {\bf 21},  3713  (1980).

\bibitem{elgin74}
R.~L. Elgin and L. Goodstein, Phys. Rev. A {\bf 9},  2657  (1974).

\bibitem{ceperley89}
D.~M. Ceperley and E.~L. Pollock, Phys. Rev. B {\bf 39},  2084  (1989).

\bibitem{cheng89}
E. Cheng, G. Ihm, and M.~W. Cole, J.\ Low Temp.\ Phys. {\bf 74},  519  (1989).

\end{thebibliography}

%-----------------------------------------------------------------------
% Method
%-----------------------------------------------------------------------

\section{Simulation method}
\label{sec:method}

Our study employs a path-integral Monte Carlo (PIMC) method for simulating
bose systems.  Details on the application 
of this method to bulk helium can be found
in the review by Ceperley,\cite{ceprev} and our modifications for the
simulation of adsorbed helium films may be found in a previous 
publication.\cite{pierce99a} We briefly review the procedure now
in order to explain the calculations presented in this paper.

PIMC evaluates properties of an $N$-body quantum system at 
the inverse temperature $\tau$ by sampling the partition function 
$Z$.  $Z$ is expanded as a path-integral 
by inserting $M$ intermediate configurations:
%%%%%%%%%%%%%%%%
\begin{eqnarray}
Z \ = & & \frac{1}{N!} 
	\sum_P \int...\int d^3 R_1 ... d^3 R_M d^3 R \nonumber \\
  & & \times \rho ({\bf R}_1,{\bf R}_2; \tau) \rho({\bf R}_2, {\bf R}_3; \tau) 
  	\ldots \rho ({\bf R}_M,P {\bf R}_1; \tau),
\label{eq:pathint}
\end{eqnarray}
%%%%%%%%%%%%%%%%
where $\rho$ is the density matrix, ${\bf R}_i$ is a configuration of 
$N$ particles, and $\tau=\beta/M$.  By taking $M$ large enough, the
density matrices at $\tau$ can be accurately approximated.  The particular
PIMC method that we employ ergodically samples both particle positions
and particle permutations.

The key element needed in the first layer simulation is an accurate
approximation for the high temperature density matrices.  For bulk
helium and helium on a smooth substrate, the starting approximation
for the density matrix can be taken with a temperature as low as
40 K.  In this paper, results for the first layer on a smooth
substrate were obtained using the same starting approximation
for the density matrix that was used in previous calculations
for the second layer.\cite{pierce99a}

However, as discussed in our previous publication,\cite{pierce99a}
a helium-graphite interaction that includes substrate corrugations 
makes the starting approximation used for smooth substrates impractical,
and so we must use a simpler form.
For these calculations,
we instead use the high temperature
expansion of the density matrix.  At sufficiently small values of the
inverse temperature, the density matrix is given by
the Trotter approximation:
%%%%%%%%%%%%%%%%%%%%
\begin{eqnarray}
\label{eq:trotter}
\exp(-\tau \hat{H}) \approx \exp(-\tau \hat{T}) \exp(-\tau \hat{V}),
\end{eqnarray}
%%%%%%%%%%%%%%%%%%%%
where $\hat{H}$, $\hat{T}$ and $\hat{V}$ are the Hamiltonian and the 
kinetic and potential energy operators, respectively, and
$\tau$ is the inverse temperature.  This allows us
to approximate the density matrices at $\tau$ as
%%%%%%%%%%%%%%%%%%%%
\begin{eqnarray}
\label{eq:hitemp}
\rho({\bf R},{\bf R}';\tau) \propto & &\exp[-\pi ({\bf R}_i
	-{\bf R}_{i+1})/\lambda_T^2 \nonumber \\
        & & -\tau (V({\bf R}_i)+V({\bf R}_{i+1}))/2],
\end{eqnarray}
%%%%%%%%%%%%%%%%%%%%
where $R$ and $R'$ are the position vectors for $N$ particle configurations, 
and $\lambda_T$ is the thermal wavelength.  This is sometimes referred
to as the semiclassical approximation, and is accurate for sufficiently
high starting temperatures.
Averaging over the potential energy terms is referred to as the
endpoint approximation.  The potential energy term $V({\bf R}_i)$ is
the sum of all helium-helium and helium-graphite interactions.
The exponent of Eq.\ \ref{eq:hitemp} is the first term in an expansion
in powers of $\tau$.\cite{imada84}

In the semiclassical calculations, we have used 200 K as the starting 
temperature, meaning that 200 inverse-temperature slices are 
required to reach 1 K.  We have verified that this temperature
is sufficiently high for the approximation to be accurate by comparing
the energy calculated at 4 K using 200 K and 320 K as the starting 
temperatures.  The energy values obtained agreed within error bars.  We have
also adopted a three-level bisection, $l=3$, for sampling the positions.
We verified that this level gave a lower energy at 4 K than
calculations using $l=2$ and $l=4$.  The acceptance rate using
$l=3$ is approximately 40\%.  We have further verified that the 
density matrix is well approximated at 200 K by Eq.\ \ref{eq:hitemp} by
including the next term in the series expansion in powers of $\tau$.  
Energy values calculated with this starting approximation agreed within error
bars with those that used Eq.\ \ref{eq:hitemp}.  

We did find that higher order terms in $\tau$ must be included in the
energy estimator.  The values we report in Sec.\ \ref{sec:sqr3} are
obtained using 
%%%%%%%%%%%%%%%%%
\begin{eqnarray}
\label{engest}
<E> = \frac{3 N}{2 \tau} + <T> + <V> + \tau^2 \hbar^2/(8 m) <(\nabla V)^2>
\end{eqnarray}
%%%%%%%%%%%%%%%%%
for the energy expectation value,
where $N$ is the number of particles, $<T>$ and $<V>$ are the kinetic and
potential energy expectation values, and $m$ is the mass of the helium
atom.  Note that in PIMC the effective action, defined as
the natural log of the density matrix, not the energy estimator, is
used to choose between configurations in the Monte Carlo procedure.

Once the machinery for the Monte Carlo method is in place, helium-helium
and helium-substrate potentials are required for input.  We use
the Aziz\cite{aziz92} potential for the helium-helium interaction.
For the helium-graphite interaction, we have adopted the 
anisotropic Lennard-Jones potential proposed by
Carlos and Cole.\cite{cole80}  This
potential can be expressed as a Fourier series in the reciprocal lattice
vectors, {\bf G}, of the graphite substrate.  In cylindrical coordinates
($\rho$,z) the expansion is
%%%%%%%%%%%%%%%%%%%%
\begin{eqnarray}
\label{eq:vgr}
V({\bf r}) = V_0(z) + \sum_{{\bf G}} V_{{\bf G}}(z) \exp(i{\bf G 
		\cdot \rho}), 
\end{eqnarray}
%%%%%%%%%%%%%%%%%%%%
where $V_0(z)$ is the laterally averaged potential, and $V_{{\bf G}}(z)$
gives the corrugation strength.  The mathematical forms for these
terms are given elsewhere, \cite{cole80} and the series
converges rapidly.  For the smooth substrate, we
use only $V_0(z)$.  For calculations that included corrugations, we
kept the six lowest, equivalent values of {\bf G} in the expansion. 

The limitation of our method is thus related to the accuracy of the
potentials available to us.  It is possible, for instance, that
the substrate may substantially mediate the helium-helium 
interaction.\cite{bruch82}
This is the so-called McLachlan interaction \cite{mclachlan64}.  We have
performed calculations both with and without this term.  Another possible
concern is the helium-graphite potential, Eq.\ \ref{eq:vgr}, which
may overestimate the corrugation strength.  Lowering the corrugation
height will effect the properties of the first layer, with the favored
phase becoming liquid instead of solid at sufficiently small corrugations.
We have examined this effect by repeating some of the calculations with
the corrugation strength set to 50 \% and to 75 \% of the value obtained from
the Carlos-Cole model.  Details of these tests of the limits of our
interaction model are given below.

Particle permutations were also included in the calculations and were
observed in the film on the smooth substrate at low densities.  
However, we have found that permutations do not play a role in the 
first layer on the corrugated substrate.  We have allowed for 
permutations at intermediate densities for temperatures as low as
0.571 K with the level of the path bisectioning taken as large as
5 (32 beads updated in one move) but never observed any particle exchanges.  
This has been checked in calculations both with and without the McLachlan
interaction.

